Author's response to reviews

Title: A theory of organizational readiness for change

Authors:

Bryan J Weiner (bryan weiner@unc.edu)

Version: 5 Date: 12 October 2009

Author's response to reviews:

Page 9 of the proofed document contained a typographical error.

The original contained the following sentence:

"The might value it because managers support it, opinion leaders support it, or peers support it."

I changed the first word from "The" to "They."

The uploaded manuscript now reads:

"They might value it because managers support it, opinion leaders support it, or peers support it."